Author’s response to reviews

Title: A botanical from the antiproliferative Cameroonian spice, Imperata cylindrica is safe at lower doses, as demonstrated by oral acute and sub-chronic toxicity screenings.

Authors:

Paul Nayim (nayim paul@yahoo.fr)
Armelle T. Mbaveng (armbatsa@yahoo.fr)
Arsene M. Ntyam (arsenemsc@gmail.com)
Victor Kuete (kuetevictor@yahoo.fr)

Version: 4 Date: 15 Jul 2020

Author’s response to reviews:

Detailed answers to reviewer and Editor

Editor Comments:
Thank you for submitting your revised manuscript. The reviewer has suggested that the absence of leaf extract and water extraction are limitations of this study. Therefore, I would be grateful if you would add a discussion of these limitations to the discussion section.

ANSWER:
Thank you, this has been added at the of the discussion section